Editor Comments: Thank you for the opportunity to further revise this manuscript. Please see below for responses to comments. I have addressed review comments in my previous revised submission.

1. Textual overlap

We note that there is some textual overlap with previously published works, especially those in Background, Results and Discussions. Please rephrase these sections in your own words to avoid direct overlap (please note, we cannot accept direct textual overlap with any previously published works even if the authors of the work are yourselves).

Please note that the software used to detect overlaps is imperfect and highlights two or more consecutive matching words, therefore, please just focus on those whole sentences/paragraphs highlighted. If this study uses methodology from a previously published work, please provide a summarizing statement in the methodology together with a citation to the original paper.

Please find the report in the attachments to aid your revisions.

I have rewritten the sections highlighted
2. Authors' contributions

Please represent authors' names using their full initials, not their full name, in the Authors’ Contributions section.

For example: ‘AB carried out the molecular genetic studies, participated in the sequence alignment and drafted the manuscript. BC participated in the design of the study, carried out immunoassays and performed the statistical analysis. CD conceived of the study, and participated in its design and coordination. All authors read and approved the final manuscript. Done with initials

3. Clean copy

Please ensure that when you upload your revised submission that it is in the final form for publication. Please remove any tracked changes or highlighting and include ONLY a single clean copy of the manuscript (and please ignore any automated requests to show tracked changes). Should you wish to respond to these revision requests, please include the information in the designated input box (Respond to Reviewers) only.

I have only provided a clean copy with revisions required due to contextual overlap as suggested by editor.